Author's response to reviews

Title: A randomised controlled trial to assess the effectiveness of a single session of nurse administered massage for short term relief of chronic non-malignant pain

Authors:

Kate Seers (kate.seers@warwick.ac.uk)
Nicola Crichton (crichtnj@lsbu.ac.uk)
June Martin (june.martin@orh.nhs.uk)
Katrina Coulson (katrina.coulson@derbyhospitals.nhs.uk)
Dawn Carroll (dawncarroll@btinternet.com)

Version: 8 Date: 22 May 2008

Author's response to reviews:

three very minor changes only - an "or" replacing "of" line 11 of methods, and extra "have been" on p7, line 8, and 1st 4 titles of references had been put in bold - none of others, so bold removed.